Author’s response to reviews

Title: High neutrophil to lymphocyte ratio confers poor prognosis in patients with small cell lung cancer

Authors:

Dan Liu (liudanscu@qq.com)
Yi Huang (huangyijy@med.uestc.edu.cn)
Lei Li (350927594@qq.com)
Juan Song (516209664@qq.com)
Li Zhang (zhangli7375@scu.edu.cn)
Weimin Li (weimi003@gmail.com)

Version: 2 Date: 03 Dec 2017

Author’s response to reviews:

Respond to reviewers:

1. Please add all authors’ email addresses before the abstract.

We have added the authors’ email addresses.

2. On uploading your revisions, please remove all tracking/highlighting and instead upload a single clean version of the manuscript.

We have removed all tracking/highlighting.

3. Please format your abstract and document according to the formatting guidelines: https://bmccancer.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

The format of manuscript was revised according to formatting guidelines.
4. Under 'Ethical approval and consent to participate' (within declarations) please add your ethical approval reference number.

The ethical approval reference number has been added.

5. Under Funding (within declarations) - please add the phrase: 'The funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript'.

We have added the phrase.

6. Please improve the resolution of your figures.

The high-resolution figures were uploaded.